Author's response to reviews

Title: Patterns of perceived barriers to medical care in older adults: a latent class analysis

Authors:

Joshua M Thorpe (jthorpe@pharmacy.wisc.edu)
Carolyn T Thorpe (cthorpe@wisc.edu)
Korey A Kennelty (knelty@wisc.edu)
Nancy Pandhi (pandhi@wisc.edu)

Version: 2 Date: 14 December 2010

Author's response to reviews: see over
December 13, 2010

Melissa Norton, MD  
Editor-in-Chief  
*BMC Health Services Research*

Dear Dr. Norton:

Please accept this brief revision of the manuscript, "Patterns of Perceived Barriers to Medical Care in Older Adults: A Latent Class Analysis," submitted for consideration for publication in the journal *BMC Health Services Research*. The original manuscript included tracked changes which should not have appeared in the manuscript. This revision has removed those tracked changes.

Thank you for your consideration and I look forward to future correspondence.

Sincerely,

Joshua M. Thorpe, MPH, PhD